Dear Editor,

Thank you very much for considering our manuscript entitled “Retroperitoneal parasitic fetus: a case report” for publication in the World Journal of Clinical Cases. We thank the editor and reviewers for providing constructive comments and suggestions, which improved the manuscript substantially.

We hope that this further revised manuscript is now acceptable for publication in your Journal.

Yours sincerely,

Jian Chen
Responses:

Reviewer #1:

Only the quality of the images should be improved, I have no special comments.

Response: We have revised the image with reference to the guidelines.

Reviewer #2:

The fetus in fetu is a rare congenital anomaly. The study reported a rare case of a male infant aged 26 days presenting with retroperitoneal parasitic fetus. The case is well described and well discussed. The reviewer suggests to accept this case report after a minor editing.

Response: Thanks for your comments.

1) *Science editor*: This case report study is very interesting. In this case report, the authors reported an rare case of retroperitoneal parasitic fetus. The case is described in detail and well discussed. Authors should updated the images, and edit the manuscript carefully.
Response: The manuscript and images have been revised appropriately.

2) Company editor-in-chief: Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Response: Original figures are provided in a PPT file. A copy of the funding document has been uploaded.